Author's response to reviews

Title: Risk Factors for Postoperative Hypoxemia in Patients Undergoing Stanford A Aortic Dissection Surgery

Authors:

Yinghua Wang (pen1019@hotmail.com)
Song Xue (xuesong64@163.com)
Hongsheng Zhu (zhuhongsheng@sina.cn)

Version: 2 Date: 5 November 2012

Author's response to reviews: see over
Dear Editor,

We would like to submit the enclosed manuscript entitled "Risk Factors for Postoperative Hypoxemia in Patients Undergoing Stanford A Aortic Dissection Surgery", which we sincerely hope to be considered for publication in your journal.

Aortic dissection is one of the most urgent medical cases for its high mortality and morbidity. Although open heart surgery for treating Stanford A aortic dissection has been a universal option, the patients are still threatened by the lethal perioperative complications such as hemorrhage, hypoxemia, renal dysfunction, cerebral dysfunction and etc. The purpose of this study is to identify the risk factors for postoperative hypoxemia in patients with Stanford A aortic dissection surgery and their relation to clinical outcomes.

The authors claim that none of the material in the paper has been published or is under consideration for publication elsewhere.

I am the author and Prof. Song Xue is the corresponding author and our address and other information is as follows,

Address: Cardiovascular Surgery Dept. RenJi Hospital, Shanghai JiaoTong University, School of Medicine.
1630. Dongfang Rd., Shanghai, P.R. China, 200127
E-mail: pen1019@hotmail.com
xuesong64@163.com
Tel: 86-21-68383724
Fax: 86-21-68383764

Thank you very much for consideration!

Yours

Sincerely

Yinghua Wang
Song Xue
Hongsheng Zhu